Author’s response to reviews

Title: Association between brain-derived neurotrophic factor and von Willebrand factor levels in patients with stable coronary artery disease

Authors:

Hong Jin (jinhong5985@163.com)
Yifei Chen (cyfwuxi@sohu.com)
Bilei Wang (wangbilei@126.com)
Yi Zhu (420948206@qq.com)
Long Chen (longchen.crown@163.com)
Genshan Ma (magenshan@hotmail.com)
Naifeng Liu (liunf2006@126.com)

Version: 4 Date: 29 Jun 2017

Author’s response to reviews:

Dear editor:

Thank you for your letter and the constructive comments concerning our manuscript. We have carefully reviewed your comments carefully and made minor change, which we hope meet with your approval. We replied to your questions or comments in details in the following text.

1. We note that Xiqiong Han has been removed from the author list and replaced by Long Chen since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

Reply: We agree with the editor. We have followed your advice and completed change of authorship request form.

2. Please ensure that the initials that you use for the Authors’ Contributions can be directly matched to the authors’ names in the author list, e.g. HJ for Hong Jin, YC for Yifei Chen etc.

Reply: We have carefully reviewed the Authors’ Contributions section, and we ensure that the initials that we use for the Authors’ Contributions can be directly matched to the authors’ names in the author list.
3. Please note that the Consent for Publication section relates to the patients’ consent, not the authors’ consent, for publication. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. This section must be included even if it is not applicable to your manuscript. If consent to publish is not applicable to your manuscript please write ‘Not Applicable’ in this section.

Reply: We have deleted the authors’ consent in the Consent for Publication, and we added ‘Not Applicable’ in this section.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We have revised our manuscript as a single, final, clean version.